Author's response to reviews

Title: Presence of anti-Toxocara canis antibodies and risk factors in children in a semirural population of Mexico.

Authors:

  camilo C Romero CRN (mvzcamilo@yahoo.com.mx)
  nadyeli N Nava NNC (ileydan_ladymac@hotmail.com)
  rafael R Heredia RHC (rafaesbirro@hotmail.com)
  linda B Bautista LBG (lin_bag@yahoo.com.mx)

Version: 4 Date: 9 January 2015

Author's response to reviews: see over
Hereby we send the changes:

**Title:** Presence of anti-Toxocara canis antibodies and risk factors in children in a semirural population of Mexico.

**Reviewer 1.**

The figure or graph of the distribution of seropositivity and age: there is significant data to introduce it into the writing.

2. From the geographical map, regions with Seroprevalences to Toxocara studies: settled location map.

![Amecameca and Chalco State of Mexico](image)
Reviewer 2.

Line 67: Write "species of Toxocara canis and Toxocara cati"

Discussion.

Lines 154-157: the value of p in the text is of 0.54 for respiratory disease and 0.87 for allergies, and then "is not statistically significant."

Line 159. "infection with this parasite"

Line 174. "Livestock farming" variable should be considered carefully, because there can be a bias. In this case, if the presence of stray dogs was observed.

Line 247. Placed in italics the journal's name.

-Bibliographies not cited in the text were removed.

Editor's comment:

I remove HIV

Dog with short hair < 1 cm)

Dog with long hair (> 1 cm)

PhD. CAMILO ROMERO NUÑEZ

AUTONOMOUS UNIVERSITY OF THE STATE OF MEXICO